

Review



Market Manipulation in Stock and Power Markets: A Study of Indicator-Based Monitoring and Regulatory Challenges

Yuna Hao¹, Behrang Vand², Benjamin Manrique Delgado³ and Simone Baldi^{4,*}

- ¹ School of Artificial Intelligence, Southeast University, Suzhou 215000, China
- ² School of Computing, Engineering & the Built Environment, Edinburgh Napier University, Edinburgh EH11 4BN, UK
- ³ SINTEF, Department of Architectural Engineering, P.O. Box 4760 Oslo, Norway
- ⁴ School of Mathematics, Southeast University, Nanjing 211189, China
- * Correspondence: s.baldi@tudelft.nl

Abstract: In recent years, algorithmic-based market manipulation in stock and power markets has considerably increased, and it is difficult to identify all such manipulation cases. This causes serious challenges for market regulators. This work highlights and lists various aspects of the monitoring of stock and power markets, using as test cases the regulatory agencies and regulatory policies in diverse regions, including Hong Kong, the United Kingdom, the United States and the European Union. Reported cases of market manipulations in the regions are examined. In order to help establish a relevant digital regulatory system, this work reviews and categorizes the indicators used to monitor the stock and power markets, and provides an in-depth analysis of the relationship between the indicators and market manipulation. This study specifically compiles a set of 10 indicators for detecting manipulation in the stock market, utilizing the perspectives of return rate, liquidity, volatility, market sentiment, closing price and firm governance. Additionally, 15 indicators are identified for detecting manipulation in the power market, utilizing the perspectives of market power (also known as pricing power or market structure), market conduct and market performance. Finally, the study elaborates on the current challenges in the regulation of stock and power markets in terms of parameter performance, data availability and technical requirements.

Keywords: market manipulation; stock markets; power markets; market monitoring; indicators; market power

1. Introduction

1.1. Reasons and Importance of Monitoring the Stock and Power Markets

The stock and power markets, as global markets with significant annual trading volumes, represent two major pillars in the market structure worldwide. Electricity is one of the most versatile sources of energy in the world, and thus, it plays an essential role in the progress and development of human society. According to a survey analysis by Statista [1], the global electricity trading volume in 2018 was 2339 billion kWh, with an increase of 2.6% compared to 2013. Fair cost allocation among users and peer-to-peer trading can help incentivize coordination between the owners of large-scale and small-scale power resources at different levels of power systems [2,3]. Meanwhile, another survey analysis by Statista [4] showed that, in the first quarter of 2021, global equity trading totaled USD 40.3 trillion, an increase of 101.6% over the same period in 2017. It should be noted that the stock market is one of the most fundamental financial markets and has been active for a long time [5–8]. Thus, as these markets deal with such a large amount of money and people, it is important to ensure that they are fairly and properly operated.

However, the open and liberalized nature of markets make it possible to face unethical attempts to gain unfair benefits through market manipulation. For this reason, any

Citation: Hao, Y.; Vand, B.; Delgado, B.M.; Baldi, S. Market Manipulation in Stock and Power Markets: A Study of Indicator-Based Monitoring and Regulatory Challenges. *Energies* **2023**, *16*, 1894. https://doi.org/10.3390/en16041894

Academic Editors: Danial Esmaeili Aliabadi and Tiago Pinto

Received: 29 December 2022 Revised: 3 February 2023 Accepted: 7 February 2023 Published: 14 February 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/licenses/by/4.0/). country hosting large-volume markets has its dedicated regulator institution, which is responsible for market regulation (in Section 2) and allows the markets to operate properly in a fair, orderly and reliable environment.

Recent years have encountered an increasing use of APIs (Application Programming Interfaces) acting on the market (both the stock market and the power market). APIs are application interfaces created to meet the increasingly large scale of software in recent years and are also a convention for the interface of different components of the software system. APIs allow different software applications to be linked together in a lightweight way, allowing quick access to data from other platforms, and also allowing third-party openers to build on top of existing products and services. According to [9], the use of APIs reached 52% of the volume on the Nord Pool power exchange's intraday markets in July 2019, compared to only 13% of total transactions using APIs in the same month in 2018. Nord Pool [10] is one of the largest electricity trading markets in the world, operating in numerous countries, such as Sweden, Norway, Finland, the UK and Estonia. More than 90% of the total electricity consumption in the Nordic region is traded through Nord Pool. Similarly, APIs occupy a high position in the stock market. According to [11], a large number of APIs are active in the stock market permanently. There is large consensus among experts that, with the rapid progress of Artificial Intelligence (AI), machine learning-based APIs may soon have a massive impact on orderly running markets and may even potentially pose a threat to equity and order in the stock and power markets. Aliabadi et al. [12] and Liang et al. [13] show that advanced pricing algorithms may, even unintentionally, create collusion cases for power-generating companies. Mizuta [14] pointed out that if an AI replaces a human as a market trader, its behavior would affect the market price and, due to its strong learning capability, it may learn how to manipulate the market.

Indeed, it seems that the development of algorithmic manipulation is already 'stimulating' the growth of market manipulation [15–17]. Cases of algorithmic manipulation derived by APIs use have been frequently reported in the stock and power markets [18]. Algorithmic trading is the pre-programming of code based on rules or developer's tendencies, which immediately execute a buy or sell object to complete a market operation when current market conditions meet pre-defined conditions [19,20]. In recent years, nearly 70% of the total trading volume in developed stock markets comes from algorithmic trading based on learning algorithms [18], such as Q-learning [21], Genetic algorithms (GA) [22] and Long Short Term Memory (LSTM) [23]. In the US, algorithmic trading accounted for approximately 60–73% of the total equity trading in 2018 [19]. According to [24], the total algorithmic trading market is expected to reach USD 18.8 billion by 2024. The advantages of algorithmic trading include accuracy in determining the current state of the market, automatic order processing, high-speed order processing [25], high-speed data retrieval and decision making, and, most importantly, the ability of trading algorithms to process orders without being influenced by human emotions, ultimately increasing profitability [24]. The advent of algorithmic trading has led to the development of High-Frequency Trading (HFT) [26]. Unfortunately, the emergence of algorithmic trading has provided powerful technical support for market manipulation [27]. There are reported case, such as in the flash crash in the eastern United States in 2010, where companies used algorithmic trading to manipulate stock market prices. Filling or cancelling large orders quickly caused stock market prices to fall sharply in a short period of time, resulting in market disruption, as well as serious financial losses (Section 3) [6].

Thus, these algorithmic trades already have a tremendous impact on today's stock and power markets, and they make market regulators face big challenges. Indeed, the complexity of market trading makes it difficult to detect different forms of manipulation, leading to distorted market trading going undetected for a long time and affecting the efficient guidance of the market for optimal resource allocation [28].

1.2. What Are the Indicators to Evaluate These Markets?

1.2.1. Stock Market

Several indicators have been proposed to measure the state of the stock market. This paper recalls and collects 10 indicators, which have been proposed to assist in monitoring the possibility of stock market manipulation over a certain period from the perspective of return rate, liquidity, volatility, market sentiment, closing price and firm governance by considering different scenarios. These indicators, which will be described in detail in Section 4.1, can help market regulators detect market participants while they attempt to manipulate the stock market and also provide evidence for judicial conviction in market manipulation cases.

1.2.2. Power Market

Similarly, several indicators have been proposed to measure the state of the power market. The monitoring of the power market can be divided into three aspects: the implementation of market rules, the identification and correction of manipulation behaviors (behaviors not stipulated by market rules), and the improvement of market rules or scheduling procedures [29]. In industrial economics, the criterion for determining effective competition is the structural-conduct-performance (SCP) analytical framework [30]. Therefore, market structure monitoring (also known as market power monitoring), market conduct monitoring and market performance monitoring are investigated to determine whether the market is manipulated. This paper summarizes 16 parameters from the 3 perspectives of market power, market conduct and market performance described in Section 4.2.

1.3. What Are the Similarities between These Two Markets?

The power market and the stock market have many aspects in common, as they both follow similar concepts. Firstly, in terms of trading venues: trading in the stock market usually takes place on a stock exchange, while trading in the power market takes place on a power exchange with similar functions to a stock exchange. Secondly, in terms of participation: in the power market, participants (e.g., generators) have a right to power into and back from the grid through the transmission network, in a similar way as investors in the stock market are free to buy and sell shares in companies. Finally, in terms of influence: the power market and the stock market are both highly liquid, and both have a significant impact on the economy. This makes it necessary for both stock and power markets to be regulated by a third-party body to avoid the damaging effects of market manipulation.

1.4. Contribution of This Work

This study conducts a literature review to compile a set of 10 indicators for detecting manipulation in the stock market and 15 indicators for detecting manipulation in the power market. The primary contribution of this research is the compilation of these indicators for use in monitoring manipulation in these markets. In collecting these indicators, a multifaceted approach is adopted. For the stock market, various perspectives are considered, including the return rate, liquidity, volatility, market sentiment, closing price and firm governance. Similarly, for the power market, multiple perspectives are taken into account, including market power (also known as pricing power or market structure), market conduct and market performance. This paper provides detailed information on the values of each indicator that may indicate potential manipulation, as well as an analysis of the strengths and limitations of each indicator. To the best of our knowledge, there is a lack of literature that offers a comprehensive examination of the regulatory landscape across multiple countries or regions for both stock and power markets, as well as a comprehensive list of indicators for monitoring manipulation, complete with an evaluation of their advantages and disadvantages.

This study is organized as follows. Section 2 provides definitions for surveillance and manipulation in the stock and power markets. It also presents an overview of the regulatory institutions that oversee these markets in different regions, including a summary of the regulatory strategies and functions in each region. Section 3 presents an analysis of reported cases of manipulation in recent years in the stock and power markets. Section 4 presents a thorough literature review of existing indicators for detecting manipulation in the stock and power markets. Section 5 examines the current challenges and limitations in data acquisition, parametric performance and technical requirements in these two markets. Finally, in Section 6, the study concludes by highlighting the current state of market surveillance and manipulation detection in the stock and power markets.

2. Surveillance in the Stock and Power Markets

This section details the definitions of market manipulation and market surveillance. As each country has a specific authority to regulate the market to prevent market manipulations, we consider the regulatory authorities in four exemplifying regions, namely, Hong Kong, the United Kingdom, the United States and the European Union. We consider both stock and power markets.

2.1. Surveillance in the Stock Market

According to [31,32], manipulation in the stock market is defined as "intentional or willful conduct designed to deceive or defraud investors by controlling or artificially affecting the price of securities, or intentional interference with the free forces of supply and demand, ultimately causing stock prices to fluctuate up and down, disrupting the order and function of the market and affecting the free and fair functioning of the stock market". Correspondingly, surveillance in stock markets refers to the use of computer algorithms or other tools that can detect market manipulation in various forms of transactions, allowing regulators to more effectively monitor market trading activities [33].

2.1.1. Hong Kong Stock Market Regulator

In Hong Kong, since the enactment of the Securities and Futures Ordinance (SFO) in 2003, the Securities and Futures Commission (SFC) [34] has become the principal authority for the regulation of the stock market in Hong Kong. Specifically, the SFC in Hong Kong is responsible for monitoring changes in the stock market daily. Once abnormal circumstances are identified, an investigation occurs on, for example, the source of the purchase or the sale of the stocks involved, which refers to the reports submitted by the securities institutions to see the background information of the investors. If any suspicion is found, a case is closely monitored and examined for more action. As an example, when faced with the outcome of market manipulation, it can be dealt with by administrative and criminal rulings, with administrative rulings made by the Market Misconduct Tribunal and criminal sanctions decided by the Hong Kong courts.

2.1.2. United Kingdom Stock Market Regulator

In the United Kingdom, the market manipulation legislation adopted the Market Abuse Regulation (MAR) No596/2014 in 2016 to prevent market manipulation. Market manipulation cases occur mainly through the Fraud Act 2006 (FA) and the Regulation on Wholesale Energy Market Integrity and Transparency (REMIT) to adjudication. The UK stock market regulator consists of the Financial Conduct Authority (FCA) [35] and the Serious Fraud Office (SFO) [36]. Between these, the FCA is the leading financial services regulator, and it was established in April 2013 to take over the responsibilities for conduct and related Prudential Regulatory Authority (PRA) affairs from the Financial Services Authority (FSA) [35]. The three main objectives of its activities are: (1) to secure a degree of protection for consumers to protect consumers, (2) to protect and strengthen the integrity of the UK financial system and (3) to promote fair, effective and reasonable competition in the interests of protecting consumers [35]. The FCA is an independent organization, monitored and controlled by the Treasury and Parliament in the UK, and financially serviced by companies and other institutions authorized by the FCA [37]. In addition, the FCA is empowered with rule making and investigative and enforcement powers [37]. Once the FCA regulates a market, it first investigates whether a company meets the minimum standards and then authorizes the company to enter the market. After the FCA has granted authorization, it continuously monitors whether the company meets the requirements. When the FCA finds that the company may harm the market and the interests of consumers, it requests relevant materials from the company for review [34,35]. As a legal enforcement agency, the FCA can penalize firms by stopping them from selling specific services and products, removing firms from the regulated industry, fining or prosecuting individuals, etc. [38,39]. The Serious Fraud Office (SFO) [32,36] is another UK regulatory body, but it is non-governmental. The SFO [40] investigates only the most serious economic crimes, such as suspected new types of fraud and situations where there is a very important public interest element, and conducts criminal proceedings for these complex economic crimes.

2.1.3. United States Stock Market Regulator

In the United States, the Securities and Exchange Commission (SEC) [41] is an independent, federal government, regulatory agency with primary responsibility for protecting all market participants and maintaining the fair and orderly functioning of the securities markets, based on the Securities Exchange Act of 1934 (SEA) and the Commodity Exchange Act (CEA) [41]. The SEC ensures that market participants are able to complete transactions in the securities market fairly and equitably, primarily by ensuring that companies make truthful representations about their business [38,39]. The SEC measures market manipulation mainly through Intent-Based Approaches, Fischel /Ross/Easterbrook Approaches, Intent and Unlawful Conduct/Harm Approaches or a combination of these approaches [42,43].

2.1.4. European Union Stock Market Regulator

In the EU, the European Securities and Markets Authority (ESMA) [44] is an independent EU body. It focuses on three main objectives that follow the Market Abuse Regulation (596/2014/EU) (MAR), namely, investor protection, orderly markets and financial stability. In order to achieve these objectives, ESMA sets market entry thresholds, regularly publishes lists of non-compliance for disciplinary purposes and adopts a risk-based approach to regulation. Once manipulation behavior is detected in the market, penalties are determined in accordance with Article 36a of the Credit Rating Agencies (CRA) Regulation and Article 25j and Article 65 of the European Market Infrastructure Regulation (EMIR) Regulation, and administrative sanctions are imposed under MAR [43–45]. Table 1 summarizes the regulatory responsibilities and penalties of the above four regional or national regulators.

2.1.5. Summary

Surveillance analysis through the stock market of these countries or regions is summarized in Table 1. It also demonstrates their regulatory responsibilities and the penalties of them.

Table 1. Regulatory measures of manipulative behavior in the stock market for Hong Kong, the United Kingdom, the United States and the European Union.

Country/ Region Regulator		Main Basis	Principles of Fines/Penalties	Regulatory Approach
Hong Kong	Securities and Futures	Securities and Futures Ordi- nance (SFO)	Administrative and criminal	Set market access principles,
	Commission		rulings (fines and fixed term	and continuously examine all
	(SFC) [34]		of imprisonment).	companies for unusual

United Kingdom	Financial Conduct Au- thority (FCA) [37] and Serious Fraud Office (SFO) [36]	- Market Abuse Regulation (No596/2014) (MAR)	Remove firms from the regu- lated industry, fine or prose- cute individuals, for example.	behavior. Conduct targeted investigations if unreasonable conditions are identified.
United States	Securities and Ex- change Commission (SEC) [41]	Securities Exchange Act of 1934 (SEA) and the Commod- ity Exchange Act (CEA)	Fines and the fixed term [41] of imprisonment.	
European Union	European Securities and Markets Author- ity (FSMA) [44]	Market Abuse Regulation (596/2014/EU) (MAR)	Fines and fixed terms of im- prisonment.	Adopted a risk-based ap- proach to its supervision.

2.2. Surveillance in the Power Market

According to Klei [28], power market manipulation is defined as "profiting by changing the price of financial assets", which means power market manipulators use various means to create an "artificial price". Moreover, according to [11,43–47], surveillance in the power market means supervising and managing the power market in accordance with national regulatory regulations, prohibiting market manipulation and any potential for insider trading, modifying imperfect market rules to prevent market manipulation, maintaining and promoting a fair and limited market, and safeguarding the rights and interests of all market participants and consumers. In the face of any breach of regulatory regulations, it is taken into account by the relevant legal requirements.

2.2.1. Hong Kong Power Market Regulator

Electricity in Hong Kong is supplied by two separate companies, the Hong Kong and China Light and Power Company Limited (CLP) and the Hongkong Electric Holdings Limited (HK Electric), depending on the geographical location. In order to strengthen the regulation of electricity, the Hong Kong Government has entered into a Scheme of Control Agreements (SCAs) [48] with each of these two companies. The SCAs set out the responsibilities of the power companies and the powers of the Hong Kong Government to regulate them. The Hong Kong Government can control the price of electricity through the Development Plans submitted by the power companies and the annual tariff reviews and audits. This is precisely conducted by the power companies, together with the maximum rates of return set for the power companies in the SCAs. Energy control in Hong Kong is mainly a responsibility of the Environment Bureau and the Electrical and Mechanical Services Department. The Environment Bureau formulates energy policies, plans energy development and regulates the economic situation of the energy market. The Electrical and Mechanical Services Department is responsible for safety matters and provides some technical support to the Environment Bureau in economic regulation.

2.2.2. United Kingdom Power Market Regulator

In the UK, the electricity regulators are the Department for Business, Energy and Industrial Strategy, the Gas and Power markets Authority, the Office of Gas and Power markets (Ofgem) and the Competition and Markets Authority [49]. Among these, Ofgem is the main one to regulate the UK electricity sector, including electricity trading [49] and market manipulation. Ofgem's key responsibilities include strict enforcement of government policy and monitoring of the power market [49], a strict crackdown on energy market manipulation [50], and imposing severe penalties on companies or individuals who manipulate the market [50]. Ofgem analyzes the presence of manipulation in the power market by testing several indicators, such as market share and concentration. The Regulation on Energy Market Integrity and Transparency (REMIT) guarantees Ofgem the right to collect information, monitor transactions and take action when Ofgem identifies a suspicious company or suspects manipulation or insider trading [51].

2.2.3. United States Power Market Regulator

In the United States, there are three institutions responsible for monitoring the electricity system, namely, the Market Monitoring Units (MMU), the ISO/RTOs and the Federal Energy Regulatory Commission (FERC) [52].

The MMU reviews market performance, assesses market efficiency and competitiveness, investigates the exercise of market power and implements mitigation measures as appropriate [52]. The MMU also monitors market anomalies, modifies and improves market rules, etc. [52]. In short, the MMU conducts regular analysis, for example, hourly or daily, depending on the market, to ensure that no company is interfering with the efficiency of the market while profiting from anti-competitive behavior. While the MMU finds any fraud throughout the power market, it co-operates with FERC and other jurisdictions to negotiate to fix the market rules [52].

In most states, the FERC has ultimate regulatory authority over the ISO/RTO and wholesale markets [52]. FERC concentrates on regulating the transmission and wholesale sale of electricity in commerce between continents, monitoring and investigating the energy markets, as well as using civil penalties and other sanctions against energy organizations and individuals who violate FERC rules in energy markets [53]. Currently, the FERC requires all participants in the power market and entities subject to regulation to provide real-time enhanced data to better monitor the market [54].

2.2.4. European Union Power Market Regulator

There is a two-tier regulatory system in the EU, i.e., the National Regulatory Authority (NRA) [55] and the European Union Agency for the Cooperation of Energy Regulators (ACER) [56]. Between them, the NRA investigates suspected cases of market abuse and enforces the Regulation on Wholesale Energy Market Integrity and Transparency (RE-MIT) and Fraud Act 2006 (FA) as a penalty within their national legal framework. As an EU Regulation, REMIT is directly effective in all EU Member States. Specific enforcement and sanctions are carried out by the NRA, while ACER's main responsibilities are: to participate in the creation of European network rules, to advise on energy-related issues in Europe, and to monitor and report on the development of the European energy market in terms of the integrity and transparency of the wholesale energy market regulations.

2.2.5. Summary

Surveillance analysis through the power market of these countries or regions is summarized in Table 2. It also demonstrates their regulatory responsibilities and penalties of them.

Country or Region	Regulator	Main Basis	Principles of Fines/Penalties	Regulatory Approach
Hong Kong	Environment Bureau	Scheme of Control Agree-	No	
	Litti formient Dureau	ments (SCAs)		
	The Office of Gas and	Regulation on Wholesale En-	Fines and criminal prosecu	
United Kingdom	Power markets	ergy Market Integrity and	tion [57]	
	(Ofgem)	Transparency (REMIT)	tion [57]	In addition to fines, take
			Impose fines of up to USD 1	measures to regulate the
	Federal Energy Regula-	Regulations of the FERC Anti-	million per day for each viola-	behavior of market mem-
United States	tory Commission	Manipulation Act and its re-	tion by a market member	bers so that they comply
	(FERC)	lated acts	(note: this maximum fine is	with the relevant laws
			adjusted for annual inflation)	and regulations
	The European Union	Fraud Act 2006 (FA) and Reg-		
E	Agency for the Coopera-	- ulation on Wholesale Energy	No or Gross	
European Union	tion of Energy Regula-	Market Integrity and Trans-	No cap on fines	
	tors (ACER); National	parency (REMIT)		

Table 2. Regulatory measures of manipulative behavior in the power market for Hong Kong, the United Kingdom, the United States and the European Union.

regulatory authorities	
(NRA)	

3. Stock Market and Power Market Manipulations

The stock and power markets have always been subject to manipulation by various means and factors. This section presents some well-known cases of stock and power market manipulations.

3.1. Stock Market Manipulation Cases, Examples

In Hong Kong, during the period 4 September 2007–17 January 2008, 3 accounts were used to complete 128 transactions in VST Holdings Limited shares [58]. Upon investigation, these transactions did not result in a change in share ownership, but invariably inflated the share price. Following an investigation into these transactions, the SEC determined that it was price manipulation using three indicators—volume, trading interval and price reflection [58]—and fined the responsible team USD 240,000 [59].

On May 2010 during Easter, a 36-minute-long lightning crash (or flash crash) occurred in the US stock market [56,59]. The flash crash had an effect on financial markets over a period of around 10 minutes, followed by a return to normal prices [60–63]. In April 2015, after a five-year investigation, the US Department of Justice filed 22 criminal charges against a British financial trader, including market manipulation [64]. Specifically, the trader placed thousands of orders for futures contracts from 2009 and then manipulated the market by modifying the code of trading software to quickly and automatically cancel these orders [65]. Back then, the major market indexes fell more than nine percent before rebounding [66], and in an instant, a trillion dollars of market value disappeared [66]. According to the investigation, the trader made more than USD 40 million profits from illegal trading between 2009 and 2015 [66]. This flash crash undermined public confidence that financial markets could operate safely [65].

In another case, in December 2020, the Securities and Exchange Commission (SEC) filed a complaint against Barton S. Ross alleging that Barton S. Ross manipulated the stock market by spreading false information across major media outlets and other platforms, resulting in an artificially inflated price of his company's stock [67]. Investigation revealed that Barton S. Ross created and disseminated false information at least 49 times over nearly 2 years, between February 2018 and January 2020, and made an illegal profit of USD 36,000 [68]. There are many other cases of market manipulation through the dissemination of false news. For example, Mark Melnick disrupted the normal functioning of the stock market in August 2021 by using his position as a live webcast host of stock trading to spread false news and was charged by the US SEC [69]. An investigation revealed that Melnick created and spread more than 100 false rumors for no apparent reason, causing the price of the subject company's stock to rise for a short period of time and making an illegal profit of USD 374,000 [69]. Ultimately, Melnick was enjoined from violating the fraud provisions of the federal securities laws and paid prejudgment interest and civil penalties in addition to the amount of his unlawful gains [70].

3.2. Power Market Manipulation Cases, Examples

The Federal Energy Regulatory Commission (FERC) received reports in 2007 of possible manipulative trading by Barclays Bank in the Western United States power market [71]. In a subsequent investigation, the FERC determined that Barclays Bank illegally manipulated the FERC-regulated electricity spot market for 655 days between November 2006 and December 2008 at the 4 most liquid trading points in the Western United States. The manipulated trading centers involved the US states of Washington, Arizona and California [28]. The manipulation roiled the power market in the Western United States by affecting not only the wholesale price of electricity in the West [72], but the final retail price for tens of millions of consumers in the Western United States. The main allegation was "index manipulation", meaning that the units were fined for trading with the knowledge that they would lose money in order to help Barclays Bank make a profit on its positions. Specifically, Barclays' traders took large monthly positions in physical electricity and then bought or sold to close them out in the open market at a fixed price, forcing the electricity price index to rise or fall in favor of the financial swap product, making a risk-free profit [24,68]. The Petition alleged that the defendants' daily trades lost an average of USD 117,404 per month during the months they were manipulated, with a total net loss from dailies trading in those months exceeding USD 4 million. Taking into account the financial position that benefited, FERC enforcement initially estimated that Barclays unfairly profited by at least USD 34.9 million and caused at least USD 139.3 million in economic losses to other market participants [72]. In 2012, the FERC required Barclays Bank Master to pay USD 435 million in civil penalties, and a total of USD 18 million in fines was imposed on 4 staff members who were primarily involved in this manipulation. Ultimately, in July 2017, the FERC and Barclays reached an agreement for Barclays to pay a USD 105 million fine [73].

Similarly, there have been instances of manipulation of the power market in the UK. In 2016, Ofgem's investigation found that InterGen had manipulated the power market by sending false "physical information" to the National Grid electricity system operator (ESO) on four days (31st October, 7th November, 8th November and 15th November 2016), resulting in InterGen manipulating the power market and profiting from it [74]. Specifically, InterGen sent "physical notices" stating that they could not produce during peak hours in order to induce ESOs to pay them for additional production and to expend the paid generation hours during the day, ultimately making a substantial amount of ill-gotten gains in the balancing mechanism [74]. The investigation found that InterGen made GBP 12.8 million in illegal profits from market manipulation [75].

According to official reports [72], Ofgem monitoring found that between March 2019 and September 2020, ESB Independent Generation Trading Limited and Carrington Power Limited manipulated market prices by submitting inaccurate amounts of minimum energy available for supply to the National Grid Electricity System Operator (NGESO), allowing the NGESO to buy more energy than was needed for the balance between supply and demand, ultimately leading to an increase in the cost of electricity purchased by consumers, to the detriment of the consumer. Subsequently, ESB Independent Generation Trading Limited and Carrington Power Limited admitted that they had inadvertently manipulated the market and took timely steps to rectify the errors. Finally, they were penalized to pay GBP 6 million to the energy redress fund to support consumers [76,77].

4. Surveillance in the Stock and Power Markets

To enable a more quantitative measure of the performance of the stock and power markets, this section details the indicators used to measure the situation of these two markets and analyzes the relationship between these indicators and market manipulation.

4.1. Indicators to Evaluate the Stock Market

Stock price manipulation can be classified as action-based manipulation, information-based manipulation or trade-based manipulation [28,78]. Among these, actionbased manipulation is market manipulation accomplished by "changing the actual or perceived value of the assets" [79]; information-based manipulation is used to manipulate the stock market by creating and disseminating false information to mislead stockholders [78], and the heart of information-based manipulation is that exploiting information asymmetry can lead to different asset returns [80]; and trade-based manipulation is where a trader manipulates stocks by buying high and selling low in a way that does not meet the general profitability requirements [78]. While trade-based manipulation is the most common type of manipulation, it is also the most difficult to detect [81].

1. Rate of Return (ROR) [82]

Return is the percentage of capital gains made in the stock market over a period of time. This indicator reflects the general trend of the share price during the period. Its formula is as follows:

$$ROR = \frac{Share Price t_1 - Share Price t_0}{Share t_0} * 100\%$$
(1)

Market manipulation increases the value of ROR [83]. Aggarwal and Wu [84] proved that stock prices rise during the manipulation and fall sharply after the manipulation ends. So, the ROR increases significantly during the stock manipulation phase and exceeds 100 percent; but after the stock is manipulated, the value of the ROR will drop sharply to a negative value. It can be argued that the ROR is a good supporting indicator for determining whether a stock market is being manipulated. However, the ROR indicator tor only considers the fluctuation of stock prices from a quantitative perspective and does not take into account other dynamic factors, such as inflation.

2. Liquidity

The liquidity of a stock market is the ease and efficiency with which shares can be bought and sold without significant impact on their price. Foucault [85] provided a theoretical account of the effect of manipulation on liquidity in 1999, noting that when stock markets are manipulated, market volatility increases and order non-execution risk decreases, which leads to conservative investment quoting strategies and ultimately wider spreads, making trading costs higher. In short, market liquidity rises significantly at the time of the manipulation, but falls significantly afterwards. On the contrary, very liquid stocks are less susceptible to manipulation. Li et al. [86] and Akinmade et al. [87] also demonstrated that the emergence of manipulative behavior (after manipulation) makes stock market transactions more expensive and less liquid.

There are a number of indicators currently used to detect stock liquidity, such as Bidask spread, Turnover ratio, etc. The following are detailed.

Bid-ask spread (BAS)

The bid-ask spread [82] measures the difference between the close of the trade asking price and the bid price. It has been identified as an important measure of information asymmetry [84], liquidity and efficiency, and it is also one of the common indicators of market liquidity [86]. According to Akinmade et al. [87], there is a negative correlation between bid-ask spread and market sales. The bid-ask spread's equation is as follows:

$$BAS_{i,t}^{k} = \frac{A_{i,t}^{k} - B_{i,t}^{k}}{Midquote_{i,t}^{k'}}$$
(2)

where $Midquote_{i,t}^{k} = \frac{A_{i,t}^{k} + B_{i,t}^{k}}{2}$, $A_{i,t}^{k}$ and $B_{i,t}^{k}$ denote the best ask price and the best bid price of *k*-th transaction for stock *i* on day *t*.

When there is information asymmetry in the market, the BAS value increases. As highlighted by Gerace et al. [82] and Aggarwal and Wu [84], market manipulation expands the value of BAS, allowing trades to take place in a context of information asymmetry. The higher the value of BAS, the less liquid the market is. However, bid-ask spreads are not sensitive to the size of transactions and do not reflect the ability of market prices to change in market volume without disruption.

Amihud illiquid measure

The Amihud illiquid measure was proposed by Amihud [88] to measure the sensitivity of share prices to trading volume. The indicator of Amihud is the ratio of stock returns to trading volume over a period of time. Its formula is as follows:

$$Amihud_{i,y} = \frac{1}{D_{i,y}} \sum_{t=1}^{D_{i,y}} \frac{|R_{i,d,y}|}{|K_{i,d,y}'},$$
(3)

where $R_{i,d,t}$ denotes the return on stock *i* on day *d* of year *y*, $VK_{i,d,y}$ represents daily volume, and $D_{i,y}$ denotes the number of days for which data are available for stock *i* in year *y*.

If a change in stock trading volume brings about violent fluctuations in the share price (sharp rises and falls), the larger the Amihud indicator, the less liquid the stock; conversely, if the change in trading volume has less impact on the change in share price, the more liquid the stock is. The advantage of this indicator is that the data required for this indicator are readily available, and it can also be used to construct a long, illiquid time series for continuous observation [88].

Turnover ratio

The turnover rate is the frequency with which a stock changes hands in the market over a given period of time. It is also an indication of the liquidity of a stock. Its specific formula is as follows:

$$Turnover_{iy} = \frac{1}{D_{iy}} \sum_{t=1}^{D_{iy}} \frac{Volume_{iy,t}}{Share_{iy,t}},$$
(4)

where $Volume_{iy,t}$ explains the number of shares traded in stock *i* on day *t* of period *y*, and *Share*_{*iy*,*t*} shows the corresponding number of outstanding shares.

The Average Daily Exceptional Turnover Rate [89] is an indicator constructed on top of the turnover rate and is used to warn whether a stock market is being manipulated. The average daily exceptional turnover of shares for the $(t_2 - t_1)$ days prior to the beginning of the manipulation was:

$$V[t_1, t_2] = \frac{1}{t_2 - t_1} \sum_{t=t_1}^{t_2} \left(Turnover_t - \frac{1}{t_2 - t_1} \sum_{p=t_2 + t_1}^{t_2} Turnover_p \right),$$
(5)

Based on a China securities market share price manipulation case, Chen and Hu [89] concluded that if a stock has a certain period of time and the average daily exceptional turnover rate suddenly increases significantly compared to the previous period without corresponding positive (or negative) news in the market, then the stock is likely to be suspected of being manipulated. This index has the best timely warning function.

The turnover rate also has some disadvantages [90]. Firstly, the turnover ratio does not take into account changes in the price of the stock, as liquidity decreases when the price of a stock continues to rise. Further, the turnover rate does not take into account other conditions of the market, such as high speculation rates. Moreover, it has been realized that using the turnover ratio alone to measure market liquidity is incomplete [90,91].

Volume

Volume [84,90] is the number of shares changing hands at a given time. Volume is an increasing function of liquidity, and as trading volume becomes larger, liquidity must also increase. A spike in trading volume over a certain period of time and a decrease after that time is a likely sign that the stock is being manipulated. The advantage of this indicator is that it is easy and straightforward to measure.

3. Volatility

Volatility [92], also called risk, is the rate at which the price of a stock increases or decreases over a period of time. It reflects the degree of uncertainty in the value of the stock. Usually, stock market traders and analysts rely on different indicators to track fluctuations in order to determine the best time to buy or sell a stock. Similarly, volatility is an important indicator of whether a stock market is being manipulated. Aggarwal and Wu [84] used US market data as a case study to figure out and demonstrate that the occurrence of manipulation increases the volatility of stocks. A stock is highly volatile if its price rises sharply or falls sharply over a certain period of time. Further, sharp fluctuations in the price of a stock over a short period of time are, to some extent, an indication that the stock is at risk of being manipulated. According to Akinmade et al. [87], volatility and

bid-ask spreads are positively correlated. For each stock, its volatility is considered to be equal to the logarithm of the proportion of daily high and low share prices, namely:

$$Volatility = Log \frac{P_{max}}{P_{min'}}$$
(6)

where P_{max} and P_{min} represent the maximum and minimum stock prices, respectively. There are a number of indicators currently used to detect stock volatility, such as the Beta coefficient, detailed as follows.

Beta coefficient

The Beta coefficient is a risk index that measures the price volatility of a stock relative to the overall market and is a tool used to assess the systematic risk of a stock. Its specific formula is as follows:

$$\beta = \frac{Covariance(R_e, R_m)}{Variance(R_m)},\tag{7}$$

where R_e denotes the return on an individual stock, and R_m denotes the return on the overall market.

If $\beta > 0$, it indicates that the stock price is positively correlated with the overall market price: as the overall market price rises, the price of the stock also rises. Conversely, if $\beta < 0$, it indicates that the stock price is negatively correlated with the overall market price: as the overall market price rises, the price of the stock falls. Normal stocks have a Beta coefficient of around 1 and are positively correlated with the market. However, when a stock is manipulated, the Beta coefficient becomes smaller or even negative, indicating that the stock price becomes less positively correlated with the market index. Lu and Chen [93] found that the Beta coefficient was significantly lower only during the period of manipulation, while the difference in Beta coefficients at pre- and post-manipulation and for non-manipulated stocks was not significant. Thus, the Beta coefficient can provide some reference in terms of stock market monitoring. However, since the Beta coefficients are calculated using historical data, they cannot help with real-time monitoring.

4. Closing-price manipulation index

The closing-price manipulation index was proposed by Comerton-Forde and Tālis J.Putniņš [94]. The coefficient takes into account the characteristics of returns, spreads, trading frequencies and return reversals, which can distinguish manipulated closing prices from those that occur in normal trading. It is formulated as follows:

$$I_{manip} = \frac{1}{1 + e^{-(a+b\Delta_{return}^{sign} + c\Delta_{reversal}^{sign} + d\Delta_{frequency}^{sign} + e\Delta_{spread}^{sign} + \varepsilon)},$$
(8)

where the coefficient indicators $(a, b, c, d, e, \varepsilon)$ can be obtained from a binary logistic regression of manipulated and non-manipulated stock days using a regression model. Where Δ_i^{sign} denotes the difference-in-variables. It is formulated as follows:

$$\Delta_i^{sign} = S_i - med_s(S_i),\tag{9}$$

where the variable *i* may include return, reversal, frequency and spread; S_i denotes the sign statistic for variable *i* on a particular stock day; and $med_s(S_i)$ denotes the median sign statistic for all other stocks.

The Closing Price Manipulation Index takes a value between 0 and 1 and represents the probability of the closing price being manipulated. It tends to increase in response to anomalies in day-end returns, trading frequencies, returns reversals and spreads. This is because abnormal day-end returns and returns reversals reflect the extent to which prices deviate from their normal levels; an increase in trading frequency can indicate that manipulation has increased trading volume. The regulator can, therefore, give priority to investigating the instances with the highest data for this index.

5. Market sentiment

The relationship between shareholder tendency and stock market performance has been carefully investigated [95–99]. Market pressure can also be expressed on the side of news containing negative sentiment [100]. The analysis of past market manipulation shows that market manipulators can twist the sentiment of individual investors by manipulating public opinion [101]. Therefore, the sentiment of individual investors and the market is also an indicator to predict whether the stock market is manipulated. The following indicators can be taken into account and used while measuring market sentiment.

Analysis

The analysis mainly refers to the evaluation of the manipulated stock by securities analysts. The normal stock evaluation is the number of positive evaluations minus the number of negative evaluations of a given stock by all securities analysts over a certain period of time. The main pillar of the analysis is whether the current share price is reasonable. The higher the number of analysts' evaluations, the less likely the stock is manipulated [15]. As a case study, applying this indicator and the indicator of the number of negative outside impacts on a stock to the Chinese stock market from 2014–2016 and using the China Securities Regulatory Commission penalty cases as a reference, Liu et al. [15] found that introducing market sentiment can be a significant marginal increment to the model, and the accuracy of a stock in the market can be improved. However, various terms play an important role in this object, for instance, misled data.

Sentiment score

The sentiment score was proposed in [102] to analyze the interplay between social media news and stock returns. Its formula is as follows:

$$Sent_d = \frac{N_d(pos) - N_d(neg)}{N_d(pos) + N_d(neut) + N_d(neg) + 3'}$$
(10)

where $N_d(neg)$, $N_d(neut)$ and $N_d(pos)$ denote the daily number of negative, neutral and positive news, respectively. $N_d(neg)$, $N_d(neut)$ and $N_d(pos)$ take the values –1, 0 and 1, respectively. The constant 3 in the denominator is Laplace corrected. Deveikyte et al. [95] demonstrated experimentally that with an increase in positive sentiment, the sentiment score increases, market volatility decreases and the market is less likely to be manipulated. However, due to the high number of social networking software, there can be incomplete data collection or false data, leading to the invalidation of this indicator.

6. Firm Governance

In terms of corporate governance, the higher the concentration of a company, the higher the probability that the stock can be manipulated. Therefore, stock market conditions can also be measured by concentration.

Concentration refers to the percentage of shares held by the top largest shareholders. Insider information trading is an important way for information to manipulate prices. The concentration of shareholding increases the likelihood of trading manipulation. The more concentrated shareholding, the greater the incentive for controlling shareholders to manipulate prices in real-time through various means.

Based on the above analysis, Table 3 summarizes the advantages and disadvantages of the above stock market monitoring system indicators.

	1 41	of billouninary of power market			
Indicator	Category	Data Required	Concern of Manipulation When the Indicator	Strengths	Weaknesses
Rate of Return (ROR)	Ex-post	Share price at different times	Rises significantly (beyond 100%)	A good supporting indicator	No consideration is given to the effect of other factors on stock prices
			Liquidity		
Bid-ask spread (BAS)	Ex-post	The best ask price and the best bid price of k -th transaction for stock <i>i</i> on day <i>t</i>	Rises significantly	Information asymmetry is taken into account	Not sensitive to the size of transactions and does not reflect the ability of market prices to change in mar- ket volume without disruption.
Amihud illiq- uid measure	Ex-post	Return on stock <i>i</i> on day d of year y, daily volume and num- ber of days for which data are available for stock <i>i</i> in year y.	Falls significantly	Can also be used to construct a long, illiquid time series for continuous observation	N.A.
Turnover Ratio	Ex-post, close to real time, Ex ante	The number of shares traded in l- stock <i>i</i> on day <i>t</i> of period <i>y</i> , the corresponding number of out- standing shares	Rises significantly (beyond aver- age)	Best timely warning effect	Does not take into account changes in the price of the stock and other conditions of the market, such as high speculation rates.
Volume	Ex-post, close to real time, Ex ante	l-The number of shares changing hands at a given time	Rises significantly or falls signifi- cantly	Easy to count, simple and straightforward	Does not take into account other conditions of the market, such as high speculation rates.
			Volatility		
Beta Coefficient	t Ex-post	Individual stock and return on the overall market	Falls below average (or even nega- tive)	Provides some reference in terms of the stock market monitoring	Cannot help with real-time monitoring

Table 3. Summary of power market monitoring indices.

Closing-price Manipulation Index	Ex-post	Return, reversal, frequency and spread	Close to 1	Takes into account manipulation for which prosecution data are not readily available and pro- vides options for more effective regulation of the market	N.A.
			Market Sentime	ent	
Analysis	Ex-post, close to real- time, Ex ante	The number of the evaluation of the manipulated stock from securities analysts	Falls significantly	Simple and intuitive	There are instances where some securities analysts are also involved in manipulation, which can make these data misleading.
Sentiment Score	Ex-post, close to real- time, Ex ante	The daily number of negative, neutral and positive news	Falls significantly	Simple and intuitive	Due to the high number of social networking soft- ware, there can be incomplete data collection or false data, leading to the invalidation of this indicator.
			Firm Governan	ce	
Concentration	Ex-post, close to real- time, Ex ante	The percentage of shares held by the top <i>i</i> largest sharehold- ers	Rises significantly	Simple and intuitive	N.A.

4.2. Indicators to Evaluate the Power Market

This section summarizes the parameters of power market monitoring from three perspectives: market power, market performance and market conduct [103].

4.2.1. Market Power

Market power [104–107], which is also known as pricing power or market structure, is the ability of a company to ultimately raise prices above the competitive level, by exploiting its unique position in the industry or by using a particular method, and to profit from this. Thus, large values of market power may raise concerns of market manipulation. Identifying or measuring the size of a company's market power can be defined by indices.

1. Herfindahl-Hirschman Index (HHI)

The Herfindahl–Hirschman Index [108] is a commonly used indicator to measure market concentration. It is expressed as the sum of the squares of the market shares of all firms in a power market. It is formulated as follows:

$$HHI = \sum_{i}^{N} (S_i)^2, \tag{11}$$

where S_i is the market share percentage of firm *i*. HHI is typically in the range of 0 to 10,000. The FERC gives specific evaluation criteria [109], with an HHI of less than 1000 indicating a competitive market; an HHI between 1000 and 1800 indicating a moderately concentrated market; and an HHI greater than or equal to 1800 indicating a highly concentrated market. Although HHI requires few parameters and is simple to calculate, it does not respond dynamically to market changes and does not include certain aspects of power supply and bidding strategies [103.

2. Market Share Index (MSI)

The Market Share Index (MSI) is the percentage of a company in the overall market share. It is presented as:

$$MSI = \frac{q_i}{\rho'}$$
(12)

where q_i is the quantity supplied by a company, and Q denotes the total volume of the market. Market share can be used for long-range studies or for near real-time screening [110]. Similar to HHI, MSI is intuitive and simple to calculate, but does not dynamically reflect market changes.

The market share index and HHI are the two commonly used concentration indices, which represent the percentage of a company's share of the overall market. The logic behind it is that the more concentrated the market, the greater the likelihood that participants exercise market power [110]. However, in the power market, a generator with a small market share may have significant market power within a local area due to blockages. Therefore, market share can hardly be used as a basis for measuring the degree of market power in the power market [111].

3. Pivotal Supplier Index (PSI)

The Pivotal Supplier Index (PSI) shows whether the residual supply is sufficient to meet market demand. It is a binary index calculated as

$$PSI = \begin{cases} 0, \ \left(\sum_{i=1}^{N} AIC_i - AC_j - \sum_{i=1}^{N} hG_i\right) \ge 0\\ 1, \ \left(\sum_{i=1}^{N} AIC_i - AC_j - \sum_{i=1}^{N} hG_i\right) < 0' \end{cases}$$
(13)

where AIC_i is the available installed capacity, AC_j is the available capacity, and hG_i denotes the hourly generation of supplier *i*. This indicator is used as a measure of how critical the target generator is [112]. When the PSI is 1, it indicates that the corresponding generator is very important in the market and has the potential to use market power. PSI

is able to capture market dynamics; however, it ignores the correlation between the behavior of market participants, such as collusive behavior [103].

4. Residual Supply Index (RSI)

The Residual Supply Index (RSI) [113] is a parameter that shows the ratio of residual supply to demand. The Residual Supply Index is a measure of the extent to which a generator's capacity meets demand, after taking into account the capacity of other generators [110]. It reflects the incentive for firms to exercise market power by retaining capacity. It is defined as follows:

$$RSI = \frac{TC - CRC}{TD},$$
(14)

where TC denotes Total Capacity, CRC denotes the Company's Relevant Capacity, and TD denotes Total Demand. When the RSI for company i in a given time period is greater than or equal to 100%, it indicates that other suppliers have the ability to meet the market demand, which indicates that company i does not have market power and only has a little influence on the market price. On the contrary, when RSI is less than 100%, other suppliers cannot meet the market demand without company i, so company i has a certain market power and influence on the market clearing price [110,114]. While this parameter does not represent the potential of a company to withhold capacity, the Return on Withholding Capacity index was presented to compensate for the shortcomings of RSI [115]. Compared to HHI and MSI, RSI can capture dynamic market parameters [103]. The two parameters, RSI and PSI, are similar, but RSI is more refined.

The RSI has had an important influence, for example, as the predictor of the power market [112,113], when it was first used by FERC as an assessment indicator to determine the market power of suppliers in the electricity market [115].

5. Return on Withholding Capacity (RWC) Index

Withholding is one of the core methods of exercising market power. In terms of withholding, there are two ways for companies to exercise market power [58–61]. The first way is through reserved capacity; by reducing the supply of energy, power generators make changes in supply and demand, which leads to higher electricity prices. This strategic retention of a portion of the capacity to produce electricity leads to an increase in profits, which can also be called Physical or Quantity Withholding. The second way is through financial withholding, i.e., increasing the supplier's bid price above the marginal cost of the generator set. This strategy is risky, as there is no guarantee of market acceptance [114]. Both of these strategies have an impact on costs and prices. RWC [115] represents the possibility of capacity withholding by the supplier. It is defined as follows:

$$RWC_{i,t} = \frac{\Delta p * (running \ capacity_{i,t}-1)}{market \ price \ t},$$
(15)

where Δp denotes the estimated value for the price premium expected by the supplier *i* for withholding one MWh capacity at time *t*. When $RWC \ge 1$, supplier *i* has a high probability of withholding capacity. However, when RWC < 1, the interpretation of RWC as an indicator is limited.

6. Residual Demand Elasticity (RDE)

Residual Demand Elasticity (RDE) measures a firm's ability to raise prices by reducing output after taking into account the demand response of buyers and the supply response of competitors [116]. It is also a method for measuring whether a company has applied market power by testing the demand curve it faces [116]. The residual demand curve is usually constructed ex-post and is calculated by subtracting from the total demand curve all the offer curves bid into the market by other participants [110]. RDE is inversely proportional to market power [116]. The smaller the value of RDE, the greater the likelihood that the corresponding company has exercised market power, and vice versa.

7. Lerner Index (Bid-Cost Margins)

The Lerner Index (LI) is also known as Bid-Cost Margins. It is used to measure the proportional deviation of the price at the firms' profit-maximizing output from a firm's marginal cost at that output [117–122]. The Lerner Index formula for firm i is

$$LI_i = \frac{p_i - MC}{p_i},\tag{16}$$

where p_i and *MC* denote the price and marginal cost, respectively, at the firm's profitmaximizing output. The LI index is the higher deviation from the marginal cost.

Moreover, it is hard for LI to determine the appropriate reference level, and it is susceptible to factors other than market power [103]. The index also does not effectively reflect the response of prices to demand [122]. Although the LI is an indicator of market power, it cannot be used as a common screening tool for market monitoring due to the lack of sufficient cost data in its calculation [114].

8. Must-Run Rate (MRR)

Must-Run Rate (MRR) [123] is the ratio of the generator's Must-Run Power to its usable capacity. Its specific formula is as follows:

$$MMR_{i} = Max(\frac{D - \sum_{j=1, i=1}^{N} q_{j}}{C_{i}}, 0),$$
(17)

where *D* indicates the overall market demand, q_j presents the successfully declared electric quantity by generator *j*, and C_i denotes the generating capacity of power companies. MRR takes values from 0 to 1. When MRR is zero, it indicates that the firm has a high probability of being substituted and has no influence on market prices. When MRR is greater than 0 and less than 1, it indicates that the company has some influence in the market. However, most companies in the market have an MMR greater than 0. Even if a company has the largest MMR after ranking, there is insufficient evidence to suggest that it has engaged in market manipulation.

4.2.2. Market Performance Monitoring

Market performance monitoring is primarily designed to monitor market inefficiencies and the potential for market abuse [103]. Market performance monitoring shares some of the same methods and indicators, such as market power monitoring and behavioral monitoring, so the different indicators and methods of market performance monitoring are highlighted here.

1. Degree of market competition

According to Dale L and William L [124], the level of market competition can be divided into four categories, which are perfect competition, monopolistic competition, oligopoly and monopoly. The degree of market competition is formulated as follows:

$$CD = \frac{P_C - MC}{P_C} * 100\%, \tag{18}$$

where P_c and MC denote the upper limit of the market price and the marginal price of the market, respectively. The more competitive the market, the less likely it is to be manipulated.

2. Output Gap

This index [111] was pioneered by Potomac, the independent monitoring agency for New York State. The Output Gap index captures the difference between the economic generation and actual output at current market prices for single, multiple or all generators. Its formula is as follows:

$$OG = EO_i - AO_{i'} \tag{19}$$

where EO_i and AO_i denote the Estimated Output (or estimated power generation) and Actual Output of firm *i*. The output gap can identify the capacity-holding behavior of generators. However, the method also has drawbacks; for example, the estimation of costs, power system coupling and unit mix in the model may differ from the actual situation.

3. Bid Sufficiency Index

The Bid Sufficiency Index [119,121] is based on previous data to measure the winning bid rate of a company. It is formulated as

l

$$BS_i = \frac{S_i}{S_{a'}} \tag{20}$$

where S_i denotes the ratio of the declared electric quantity to the transacted electric quantity of the power generation company *i*, and S_a indicates the ratio of the average declared electric quantity to the transacted electric quantity across the market. This parameter provides an intuitive way of detecting the winning rate of a company over a period of time. In a free market, the higher value, the more likely it is that the company in question is involved in or dominates market manipulation.

4.2.3. Market Conduct Monitoring

Market behavior is the strategy adopted by a company to achieve higher profits and market share. The market strategy consists mainly of a price strategy and a supply strategy for generators. Due to its characteristics, conduct monitoring has the same monitoring indicators as market power monitoring and market performance monitoring, such as HHI, PSI and RSI. However, some indicators are peculiar for conduct monitoring, which are presented in the following.

Highest market price index

The highest market price index [112] is the ratio of the electricity amount when the generation company's offer was closest to the maximum price to all declared electricity amounts over a period of time. Its formulation is:

$$CP_i = \frac{q_{ai}}{q_{a'}} \tag{21}$$

where q_{ai} denotes the amount of electricity that corresponds the closest to the highest price quoted by generation company *i*, and Q_a is the amount of all declared electricity. The purpose of market manipulation is to gain more profit. Therefore, the index can be used as an auxiliary discriminator when judging whether a company is involved in market manipulation. The index is derived and ranked for all companies involved in the bidding process. The higher the value and the higher the ranking of the company surveyed, the more likely it is that the company has engaged in market manipulation.

2. Maximum price winning rate

Maximum price winning rate [108,122,125] refers to the probability of a successful bid at the highest price. Its formula is as follows:

$$CR = \frac{T_{hi}}{T_b'}$$
(22)

where T_{hi} represents the number of cycles where the market clearing price is equal to the maximum price, and T_b denotes the total amount of bidding time cycles. The larger the value, the more likely it is that manipulation is involved. Therefore, the index can be used as an auxiliary discriminator when judging whether a company is involved in market manipulation.

3. Price markup index

The price markup index [112,126] is an index primarily used for measuring the difference between the bid price and the marginal cost. The formula is as follows:

1

$$PA_i = \frac{p_{mi} - MC_i}{MC_i},\tag{23}$$

where p_{mi} and MC_i represent the bid price and marginal cost of generator *i*, respectively. The PA index responds to market profitability. This index indicates the bid characteristics or strategy of the power generator. When this indicator has a high value after ranking with other companies, it is an offensive bid strategy; when this value is less than zero, it indicates that the company would rather lose money than complete the bid. This index can be used as an auxiliary indicator to study whether a company is involved in a manipulation case. However, this parameter faces the problem that the estimated marginal costs are not always very accurate, which leads to doubts about the accuracy of the results of the price markup index [111].

4. The bid price of cleared energy monitoring index

This index was invented by Kasperowicz et al. [123] for the specific situation of the data available in the Spanish electricity system. It shows the relationship between the weighted average price of clean energy at a certain time for a particular technology and the weighted average price at the same time for all technologies. It is calculated as follows:

$$BPceM = \frac{w_{PcE_{i,t}} - w_{PcE_{t}}}{w_{PcE_{t}}},$$
(24)

where $wPcE_{i,t}$ represents the weighted average price of firm *i* at time *t*, and $wPcE_{j,t}$ represents the weighted average price of all firms at time *t*. When the value of the index is greater than zero, the corresponding firm may have triggered an increase in the final price. Conversely, if the value of the index is less than zero, the surveyed firm may have caused a decrease in the final price. Once the value is almost zero, it means that the firm's bid price has almost no impact on the overall market price. This index can be used to detect relationships between those generators that have a large impact on the final price and all generators. This indicator helps to quickly locate companies that cause changes in energy prices.

Based on the above analysis, Table 4 summarizes the advantages and disadvantages of the above power market monitoring system indicators.

Indicator	Category	Data Required	Concern of Manipulation When the Indicator	Strengths	Weaknesses
			Market power		
Herfindahl Hirschm Index (HHI)	anUsually ex ante -	Total volume of market Quantity supplied by a company	Rises significantly (beyond 1800)	- Small amount of data re- quired for the calculation - - Simple to understand -	 Does not respond dynamically to market changes Does not include certain aspects of power supply and bidding strategies
Market Share Ind (MSI)	exUsually ex ante -	Quantity supplied by a company Total volume of the market	Rises significantly	- Small amount of data re- quired for the calculation - Simple to understand	 Does not respond dynamically to market changes Does not include certain aspects of power supply and bidding strategies
Pivotal Supplier Index (PSI)	Ex-ante, close- to real-time,- ex-post -	Available installed capacity Available capacity Hourly generation of supplier i	Close to 1	- Can track dynamically changing markets.	- Ignores elasticities and market contestability fac- tors.
Residual Supply Ind (RSI)	Ex-ante, close- ex to real-time,-Com ex-post -	Total Capacity pany i's Relevant Capacity Total Demand	Falls significantly below 100%	- Can track dynamically changing markets. - Widely used	- Ignores elasticities and market contestability fac- tors.
Return on Withholdin Capacity (RWC)	ngEx-ante, ex post -	Running capacity Market price	Rises significantly beyond 1	- Can capture dynamic market parameters	N.A.
Residual Demand Ela ticity (RDE)	Ex-post	Total demand Other participants' bid	Very small	- Consider other participants' bids and market needs	- Requires bid data
Lerner Index (Bid Co Margins)	Ex ante, close ost to real time, ex-post maxim	and marginal cost at the firm profit nizing output	t-Rises significantly (but could be due t factors other than market power)	Difficulties in determining costs or appropriate competitive 'reference' levels	 Does not effectively reflect the response of prices to demand The lack of sufficient cost data in its calculation
Must-Run Rate (MRR)	- Ex-post tity b - comp	Overall market demand, Successfully declared electric quar y generator j The generating capacity of powe anies	Rises close to 1	- Intuitively	-Results are not convincing
			Market performance monitor	ing	
Degree of market con petition	m-Ex-ante, ex post -	Upper limit of market price Marginal price of a market	Falls significantly	- Intuitively	N.A.

Table 4. Summary of power market monitoring indices [110].

Output Gap	Ex-ante	-	Estimated power generation Actual output of firm i	Value rises significantly (but could b due to other factors)	e- If calculated accurately, can identify the capacity-holding behav- ior of generators	If the data used in the calculation are inaccurate, result is not credible either
Bid Sufficiency Index	Ex-post	- transa eratio - quant the m	Ratio of declared electric quantity to acted electric quantity of the power gen n company I Ratio of average declared electric ity to transacted electric quantity across arket	o - Rises significantly s	- Intuitively N.2	А.
				Market Conduct Monitorin	g	
Highest market price in dex	ⁿ⁻ Ex-post	- the clo compa -	the amount of electricity that correspond sest to the highest price quoted by generation ny <i>i</i> , the amount of all declared electricity	s ⁿ High value and high ranking	- Can assist in determining whether a company is involved inN.2 market manipulation	Α.
Maximum price winnir rate	^{ig} Ex-post	- ket cle price -	The number of cycles where the mar earing price is equal to the maximum Total amount of bidding time cycles	- Value rises significantly	- Can assist in determining whether a company is involved inN.2 market manipulation	Α.
Price markup index	Ex-post	-	Bid price Marginal cost	High value and high ranking (bu could be due to other factors)	- Can assist in determining it whether a company is involved in market manipulation	inaccurate estimate of marginal costs would make the ults of this parameter infeasible
Bid price of cleared energy monitoring index	ⁿ⁻ Ex-post	- -	Weighted average price of a firm Weighted average price of all firms	Large positive values or large negativ values	- Data available (in the Span- ish market) - Help to quickly locate com-N. panies that cause changes in energy prices	Α.

5. Existing Barriers to Stock Market Surveillance and Power Market Surveillance

Many pressing issues still need to be resolved in the stock and power markets, and this section focuses on the barriers encountered in both markets.

5.1. Barriers to Stock Market Surveillance

First, the primary challenge related to technological developments is the issue of data collection. Different stock exchanges differ in the time they capture information, the breadth of information they receive, and even the extent to which they capture such information. However, all stock exchanges share the challenge that the speed of trading on all exchanges has increased significantly, requiring regulators to adapt quickly to market developments, steadily increasing their ability to collect and process the increased volume of trading data. In the face of a complex market environment and advanced technological challenges, the ability to collect complete data and information in an accurate and timely manner is the biggest challenge for regulators [127].

Second, another challenge is to extract key and true information from the vast amount of data and information available [127,128]. The increasing amount of "trading noise" due to the advent of HFT and fully automated trading programs makes it a challenge to distinguish between real orders and trading and manipulative activity [125,126]. Algorithmic trading is bound to leave a trail, as long as it exists (e.g., executed trades, program code, etc.), so it is relatively easy to detect algorithm-based market manipulation [129]. However, when the market becomes an algorithm-driven market, it becomes difficult to accurately detect algorithm-based market manipulation. This is because interpreting the data and deducing the problematic orders from the vast amount of data available takes a lot of time and money, and requires sophisticated technology and skilled personnel. This creates a huge hurdle for regulators [129]. In addition, some companies have inefficient disclosure, which can be lagging or inconsistent, and in some cases, even evasive. This leads to the need for regulators to not only extract key information from the vast amount of information available, but also to identify unusual or false information [15].

Third, most indicators are biased to some extent, which can be also a challenge. Although most trading platforms around the world have automated monitoring systems that take into account multiple measurement factors, essentially all indicators (such as those mentioned in Section 4.1) can be somewhat biased because the indicator itself has some shortcomings that can affect the accuracy of the monitoring system to some extent. This is a barrier that needs to be addressed by regulators.

Fourth, as known from experience, most manipulation cases do not involve an isolated transaction, but usually involve one or more market participants and a series of market behaviors over a long period of time. These participants and these abnormal market behaviors ultimately lead to the manipulation of the market. This requires the regulator to look beyond a single isolated transaction and to take a broader view and a larger time horizon, integrating market information while focusing on the timing and sequence of market transactions. Accurate analysis of data and keen market monitoring skills are also challenging for agencies and their policies.

Finally, keeping up with market changes promptly is a constant challenge for market regulators. With the development of the stock market, the manipulation techniques and manipulation tools in the stock market have been upgraded, so the relevant policies and regulatory tools also need to be improved and strengthened in line with the development of the market, so as to respond to the gaps in market regulation in a timely manner.

5.2. Barriers to Power Market Surveillance

Firstly, the primary barrier to power market surveillance, as with the stock market, is the issue of timely data collection. The public release of data submitted to the system (e.g., bids, output levels, etc.) and generated in the system can increase the transparency

of wholesale power markets to some extent [130]. However, this is not always the case; for example, in the US, there is a six-month delay in the release of relevant information, and the identity of market participants is kept anonymous. This creates a significant barrier to effective market regulation, as the availability of complete and timely information and data is a prerequisite for efficient and timely regulation by the regulator and provides a good environment for effective monitoring by other market participants.

Second, data accessibility can be a challenge for regulators. For the generation side, assessment of the market power of generators is often difficult. Often, indirect parameters are needed to measure a firm's market power (Section 4.2.1 for details), but the parameters are usually obtained with the help of other parameters, and the values of these parameters are usually obtained with delays or inaccurately, or are even not accessible, which leads directly to an inaccurate assessment of market power. For example, generation withhold-ing capacity can influence market power, but obtaining generation withholding capacity is complicated [131].

Third, after systematically analyzing the economic parameters of the power market, Kasperowicz et al. [103] stated in 2016 that the lack of available source data makes the innovation of new monitoring indices and methods a fundamental challenge and an objective of power system monitoring. Similar to the indicators for the stock market, the indicators for the power market (Section 4.2) are also biased and depend on other values to be obtained when solving for them, and these values are often slow to update or even have spurious values. Errors in indicator data can mislead the regulator's analytical results. This also creates barriers to effective market surveillance.

Fourth, as with the stock market, another obstacle to monitoring the power market is extracting anomalies from the vast amount of data and using this information as a clue to uncover and verify whether a company or companies are involved in market manipulation. The extensive use of APIs in the power market, while facilitating, for example, bidding by generating companies, also creates a barrier for regulators to identify potential anomalies from the vast amount of normal quotes and orders.

Fifth, as with the stock markets, past manipulation cases have shown that the power market can be manipulated in a variety of ways, also involving multiple companies and over multiple time periods. This requires the regulator to be able to spot the manipulator's purpose hidden behind the transactions from the complexity of the transactions and to identify anomalies keenly with a macro view and senior experience.

Finally, the ability of regulators to consistently and steadily keep pace with the market is an enduring test for regulators. In total, 24% of people at the 2019 Navigating Change—Regulation and Tomorrow's Grid seminar in Brisbane felt that in the energy sector, regulation has failed to keep pace with the market [132].

According to the abovementioned literature analysis, it can be summarized that both markets are facing almost the same barriers, which can be itemized as:

- 1. how to collect and complete data and information in an accurate and timely manner,
- 2. how to extract true and key information from vast data,
- 3. how to find better indicators or overcome the shortcomings of biased indicators (how to match barriers to an aimed market),
- 4. how to accurately analyze data and improve the accuracy of algorithms, and
- 5. how to ensure that the algorithm is capable of continuous learning so that it can respond to changes in the market and upgrades in manipulation techniques.

6. Conclusions

Both stock and power markets play major roles in trading worldwide. The nature of these markets allows market manipulation and unfair strategies. It requires careful monitoring and assessment of markets to prevent any manipulations.

According to a literature review, this paper studied, reviewed and listed the existing stock and power market indicators to monitor markets and detect false signals. At present,

there are visible challenges to monitoring both the stock and power markets. Based on the comprehensive investigation, it is vital to develop clearer and more oriented data collection mechanisms and a monitoring system that can incorporate expert mechanisms and automatic learning.

This paper introduced the regulatory authorities and their responsibilities for overseeing market manipulation cases in the stock and power markets in Hong Kong, the United Kingdom, the United States and the European Union, and elaborated on six highimpact market manipulation cases. It is found that manipulation cases are generally characterized by high concealment, a large number of manipulators, a high number of manipulations and a long investigation time. Although all of these regions have set up special regulators to prevent market manipulation and have established tougher penalties based on policies, it is clear that manipulation is still common.

In order to better establish a relevant digital regulatory system, we categorize and list the indicators used to monitor the stock and power markets, and provide an in-depth analysis of the relationship between all the indicators involved and market manipulation. Through an extensive and in-depth literature reading, the framework based on market power, market performance and market behavior presented in this paper could be very important for the construction of a completely new monitoring system in the electricity market. In addition, the existing indicators for monitoring electricity markets generally have some shortcomings. A better index may also be needed for the ever-changing power markets, and there is still an urgent need to develop better data collection mechanisms and a monitoring system that can incorporate expert mechanisms and automatic learning to free staff from cumbersome and time-consuming tasks.

Author Contributions: Conceptualization, B.V. and B.M.D.; methodology, Y.H.; software, Y.H.; validation, B.V., B.M.D. and S.B.; formal analysis, Y.H.; investigation, Y.H.; resources, S.B.; data curation, Y.H.; writing—original draft preparation, Y.H., B.V., B.M.D. and S.B.; writing—review and editing, Y.H., B.V., B.M.D. and S.B.; visualization, Y.H.; supervision, B.V., B.M.D. and S.B.; project administration, B.V., B.M.D. and S.B.; funding acquisition, B.V., B.M.D. and S.B. All authors have read and agreed to the published version of the manuscript.

Funding: The first and fourth authors were supported by the Research Fund for International Scientists under Grant 62150610499 and by the Key Intergovernmental Special Fund of the National Key Research and Development Program under Grant 2021YFE098700. The second author was supported by a personal grant from The Finnish Foundation for Technology Promotion/The Foundations' Post Doc Pool.

Data Availability Statement: no new data were created.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Net Consumption of Electricity Worldwide in Select Years from 1980 to 2018. Available online: https://www.statista.com/statistics/280704/world-power-consumption/#statisticContainer (accessed on 4 January 2022).
- Morstyn, T.; Teytelboym, A.; Mcculloch, M.D. Bilateral Contract Networks for Peer-to-Peer Energy Trading. *IEEE Trans. Smart Grid* 2019, 10, 2026–2035. https://doi.org/10.1109/TSG.2017.2786668.
- Bashian, A.; Attar, T.S.; Javidi, M.H.; Hojat, M. Determination of Tariff for Wheeling Contracts Considering Fairness Congestion Cost Allocation. In *Power and Energy Systems*; ACTAPRESS: Crete, Greece, 2011.
- 4. Value of Global Equity Trading Worldwide from 1st Quarter 2017 to 2nd Quarter 2021. Available online: https://www.statista.com/statistics/242745/volume-of-global-equity-trading/ (accessed on 8 January 2022).
- Parker, K.; Fry, R. More than Half of U.S. Households Have Some Investment in the Stock Market. Available online: https://www.pewresearch.org/fact-tank/2020/03/25/more-than-half-of-u-s-households-have-some-investment-in-the-stockmarket/ (accessed on 8 February 2022).
- Share of Adults Investing Money in the Stock Market in the United States from 1999 to 2020. Available online: https://www.statista.com/statistics/270034/percentage-of-us-adults-to-have-money-invested-in-the-stock-market/ (accessed on 8 February 2022).

- 7. Charlie, B. Investment Statistics: What Percentage of the UK Population Invests in the Stock Market? Available online: https://www.finder.com/uk/investment-statistics (accessed on 9 May 2022).
- 8. Fey, G. Shareholder Numbers 2020: Germans Are Becoming More and More Enthusiastic about Stocks. Available online: https://www.dai.de/en/shareholder-numbers/ (accessed on 7 May 2022).
- 9. Algorithm Trades Hit Record on Nord Pool Intraday Power Market. Available online: https://www.reuters.com/article/europe-powertrading-idINL8N24X5V0 (accessed on 9 March 2022).
- 10. Nord Pool Home Page. Available online: https://www.nordpoolgroup.com/ (accessed on 23 April 2022).
- 11. Kingaby, S.A. The Stock Market API. In *Data-Driven Alexa Skills*; Apress: Berkeley, CA, USA, 2022; pp. 387–404. ISBN 978-1-4842-7448-4.
- Esmaeili Aliabadi, D.; Chan, K. The Emerging Threat of Artificial Intelligence on Competition in Liberalized Electricity Markets: A Deep Q-Network Approach. *Appl. Energy* 2022, 325, 119813. https://doi.org/10.1016/j.apenergy.2022.119813.
- Liang, Y.; Guo, C.; Ding, Z.; Hua, H. Agent-Based Modeling in Electricity Market Using Deep Deterministic Policy Gradient Algorithm. *IEEE Trans. Power Syst.* 2020, 35, 4180–4192. https://doi.org/10.1109/TPWRS.2020.2999536.
- Mizuta, T. Can an AI Perform Market Manipulation at Its Own Discretion? A Genetic Algorithm Learns in an Artificial Market Simulation. In Proceedings of the 2020 IEEE Symposium Series on Computational Intelligence (SSCI), Canberra, ACT, Australia, 1–4 December 2020; pp. 407–412.
- Liu, Q.; Wang, C.; Zhang, P.; Zheng, K. Detecting Stock Market Manipulation via Machine Learning: Evidence from China Securities Regulatory Commission Punishment Cases. *Int. Rev. Financ. Anal.* 2021, 78, 101887. https://doi.org/10.1016/j.irfa.2021.101887.
- 16. Susser, D.; Roessler, B.; Nissenbaum, H. Technology, Autonomy, and Manipulation. *Internet Policy Rev.* 2019, 8. https://doi.org/10.14763/2019.2.1410.
- Chiranjivi, C. Zerodha Founder, Others Share Concerns as Sebi Looks to Regulate API Trading. Available online: https://economictimes.indiatimes.com/markets/stocks/news/zerodha-founder-others-share-concerns-as-sebi-looks-to-regulate-api-trading/articleshow/88204877.cms (accessed on 7 May 2022).
- RAVI, K. Why Algorithmic Trading Is Dangerous. Available online: https://asiatimes.com/2019/05/why-algorithmic-trading-isdangerous/ (accessed on 14 December 2021).
- 19. Algorithmic Trading Market- Growth Trends, COVID-19 Impact, and Forecasts (2021–2026). Available online: https://www.mordorintelligence.com/industry-reports/algorithmic-trading-market (accessed on 14 December 2021).
- 20. James, C. Algorithmic Trading. Available online: https://www.investopedia.com/terms/a/algorithmictrading.asp (accessed on 14 December 2021).
- 21. Chakole, J.B.; Kolhe, M.S.; Mahapurush, G.D.; Yadav, A.; Kurhekar, M.P. A Q-Learning Agent for Automated Trading in Equity Stock Markets. *Expert Syst. Appl.* 2021, 163, 113761. https://doi.org/10.1016/j.eswa.2020.113761.
- 22. Straßburg, J.; Gonzàlez-Martel, C.; Alexandrov, V. Parallel Genetic Algorithms for Stock Market Trading Rules. *Procedia Comput. Sci.* 2012, *9*, 1306–1313. https://doi.org/10.1016/j.procs.2012.04.143.
- 23. Banik, S.; Sharma, N.; Mangla, M.; Mohanty, S.N.; Shitharth, S. LSTM Based Decision Support System for Swing Trading in Stock Market. *Knowl.-Based Syst.* 2022, 239, 107994. https://doi.org/10.1016/j.knosys.2021.107994.
- Erik, V.; Gabriel, P.; John, A.; Keith, M.; Darren, M. Algorithmic Trading: HPC & AI Reference Guide. Available online: https://www.delltechnologies.com/asset/en-us/products/ready-solutions/industry-market/hpc-ai-algorithmic-tradingguide.pdf (accessed on 14 December 2021).
- 25. Biais, B.; Foucault, T.; Moinas, S. Equilibrium High-Frequency Trading. SSRN J. 2012. https://doi.org/10.2139/ssrn.2024360.
- 26. High-Frequency Trading (HFT). Available online: https://www.investopedia.com/terms/h/high-frequency-trading.asp (accessed on 14 December 2021).
- 27. Lin, T.C.W. The New Market Manipulation. *Emory LJ* 2017, 66, 62.
- 28. Kleit, A.N. What Is Market Manipulation? Lessons from Barclays and Powhatan. *Electr. J.* 2019, 32, 1–5. https://doi.org/10.1016/j.tej.2019.01.001.
- 29. Bai, M.; He, A. Market monitoring and regulation in the U.S. electricity market. Price Theory Pract. 2017, 15–19.
- 30. Glenys, F. Industrial Economics; Industrial Economics Press, 2002; ISBN 978-0-8147-2625-9.
- 31. Jeffrey, M.G. What Is Stock Market Manipulation? Available online: https://www.thebalance.com/stock-market-manipulation-5184361 (accessed on 23 November 2021).
- 32. Li, A.; Wu, J.; Liu, Z. Market Manipulation Detection Based on Classification Methods. *Procedia Comput. Sci.* 2017, 122, 788–795. https://doi.org/10.1016/j.procs.2017.11.438.
- 33. Cumming, D.; Dannhauser, R.; Johan, S. Financial Market Misconduct and Agency Conflicts: A Synthesis and Future Directions. *J. Corp. Financ.* **2015**, *34*, 150–168. https://doi.org/10.1016/j.jcorpfin.2015.07.016.
- Securities and Futures Commission. Available online: https://www.sfc.hk/en/About-the-SFC/Our-role/How-we-function (accessed on 25 January 2022).
- 35. About the FCA. Available online: https://www.fca.org.uk/about/the-fca (accessed on 13 December 2021).
- 36. Serious Fraud Office (United Kingdom). Available online: https://en.wikipedia.org/wiki/Serious_Fraud_Office_(United_Kingdom) (accessed on 8 February 2022).
- 37. Wi, K. Financial Conduct Authority (UK) (FCA). Available online: https://www.investopedia.com/terms/f/financial-conductauthority-uk-fca.asp (accessed on 13 December 2021).

- 38. FCA Mission: Approach to Supervision. Available online: https://www.fca.org.uk/publication/corporate/our-approach-super-vision-final-report-feedback-statement.pdf (accessed on 1 August 2022).
- Financial Conduct Authority-Supervision. Available online: https://www.fca.org.uk/about/supervision (accessed on 13 December 2021).
- 40. Serious Fraud Office. Available online: https://www.sfo.gov.uk/about-us/ (accessed on 8 February 2022).
- 41. James, C. Securities and Exchange Commission (SEC). Available online: https://www.investopedia.com/terms/s/sec.asp (accessed on 14 December 2021).
- 42. Drakeford, C. Market Manipulation: Definitional Approaches. Available online: https://projects.iq.harvard.edu/files/financialregulation/files/market_manipulation_case_study.pdf (accessed on 14 December 2021).
- 43. Securities And Exchange Commission—What We Do. Available online: https://www.sec.gov/about/what-we-do (accessed on 14 December 2021).
- 44. European Securities and Markets Authority (ESMA) Home Page. Available online: https://www.esma.europa.eu/ (accessed on 5 April 2022).
- 46. Surveillance. Available online: https://en.wikipedia.org/wiki/Surveillance (accessed on 16 November 2021).
- 47. Nord Pool Market Surveillance. Available online: https://www.nordpoolgroup.com/trading/Market-surveillance/ (accessed on 16 November 2021).
- Hong Kong Scheme of Control Agreements. Available online: https://www.enb.gov.hk/sc/resources_publications/agreement/index.html (accessed on 26 April 2022).
- 49. Kirsti, M.; James, E. Electricity Regulation in the United Kingdom: Overview. Available online: https://uk.practicallaw.thomsonreuters.com/w-029-0803?transitionType=Default&contextData=(sc.Default)&firstPage=true (accessed on 12 March 2022).
- Jamieson, J.; Njegovan, N. Testing Energy Market Manipulation in Great Britain. Available online: https://www.jamsadr.com/files/uploads/documents/kaiser-energy-market-manipulation-a-new-regulatory-regime.pdf (accessed on 25 November 2021).
- 51. Summary of Hearing with Ofgem. Available online: https://assets.publishing.service.gov.uk/media/54bf95f540f0b6158a000010/Summary_of_hearing_with_Ofgem.pdf (accessed on 11 April 2022).
- 52. Dehdashti, E. Monitoring and Surveillance of Wholesale Electricity Markets -Roles, Responsibilities and Challenges. In Proceedings of the IEEE Power Engineering Society General Meeting, San Francisco, CA, USA, 12–16 June 2005; pp. 1971–1977.
- 53. What FERC Does. Available online: https://www.ferc.gov/ (accessed on 28 August 2022).
- 54. Davis, T.C. FERC Enhances Its Market Surveillance Tools. Electr. J. 2013, 26, 45-51. https://doi.org/10.1016/j.tej.2012.11.013.
- 55. National Regulatory Authorities (NRAs). Available online: https://www.acer.europa.eu/remit/cooperation/national-regulatory-authorities (accessed on 10 May 2022).
- 56. The European Union Agency for the Cooperation of Energy Regulators (ACER). Available online: https://www.acer.europa.eu/ (accessed on 10 May 2022).
- Sheldon, B. Enforcement Guidelines. Available online: https://www.ofgem.gov.uk/publications/enforcement-guidelines (accessed on 1 March 2022).
- 58. Gerace, D.; Chew, C.; Whittaker, C.; Mazzola, P. Judicial Determination of Stock Market Manipulation Cases in Hong Kong and Its Implications. *Australasian Accounting, Business and Finance Journal* **2014**, 8.
- 59. Gorman, T. This Week in Securities Litigation (Week Ending 26 October 2012). Available online: https://www.secactions.com/this-week-in-securities-litigation-week-ending-october-26-2012/ (accessed on 25 January 2022).
- 60. Aquilina, M.; Eyles, B.; Shao, J.; Ysusi, C. How Do Participants Behave during Flash Events? Evidence from the UK Equity Market. Available online: https://www.fca.org.uk/publication/research/research-note-participants-behave-flash-events-evidence.pdf (accessed on 30 November 2021).
- Menkveld, A.J.; Yueshen, B.Z. The Flash Crash: A Cautionary Tale About Highly Fragmented Markets. *Manag. Sci.* 2019, 65, 4470–4488. https://doi.org/10.1287/mnsc.2018.3040.
- Kirilenko, A.A.; Kyle, A.S.; Samadi, M.; Tuzun, T. The Flash Crash: The Impact of High Frequency Trading on an Electronic Market. SSRN J. 2011. https://doi.org/10.2139/ssrn.1686004.
- 63. Easley, D.; O'Hara, M. The Microstructure of the "Flash Crash". Available online: https://jpm.pm-research.com/content/37/2 (accessed on 30 November 2021).
- 64. Brush, Silla; Schoenberg, Tom; Ring, Suzi How a Mystery Trader with an Algorithm May Have Caused the Flash Crash. Available online: https://www.bloomberg.com/news/articles/2015-04-22/mystery-trader-armed-with-algorithms-rewrites-flashcrash-story (accessed on 30 November 2021).
- 65. Preliminary Findings Regarding the Market Events of 6 May 2010. Available online: https://www.sec.gov/sec-cftc-prelimreport.pdf (accessed on 30 November 2021).
- 66. Flash Crash. 2010. Available online: https://en.wikipedia.org/wiki/2010_flash_crash#cite_note-74 (accessed on 30 November 2021).
- 67. SEC Charges Former Day Trader with Market Manipulation Scheme. Available online: https://www.sec.gov/litigation/litre-leases/2020/lr24989.htm (accessed on 3 December 2021).

- SEC Complaint about Barton, S. Ross. Available online: https://www.sec.gov/litigation/complaints/2020/comp24989.pdf (accessed on 28 August 2022).
- 69. SEC Charges Webcast Host for Role in Market Manipulation Scheme. Available online: https://www.sec.gov/news/press-release/2021-206 (accessed on 3 December 2021).
- Complaint—Mark, J. Melnick—SEC.Gov. Available online: https://www.sec.gov/litigation/complaints/2021/comp-pr2021-206.pdf (accessed on 1 August 2022).
- Hale, M. FERC Anti-Manipulation Enforcement and the Barclays Proceeding: What Factors Should Regulated Entities Consider before Deciding to Follow Barclays' Path to Federal Court? Wash. Lee J. Energy Clim. Environ. 2014, 6, 48.
- 20150520 FERC vs Barclays Defendents Motions to Dismiss and Transfer. Available online: https://cms.ferc.gov/sites/de-fault/files/2020-11/20150520%20FERC%20vs%20Barclays%20Defendents%20Motions%20to%20Dismiss%20and%20Trans-fer.pdf (accessed on 28 August 2022).
- LaFleur, C.A.; Powelson, R.F.; Levine, K. United States of America Federal Energy Regulatory Commission. 2009, Available online: https://cms.ferc.gov/sites/default/files/2020-11/20171107%20Barclays%20Settlement%20Agreement.pdf (accessed on 22 November 2021).
- 74. Final Notice Regarding the Imposition of a Financial Penalty under Regulation 38 (1) and 38 (5) of the Electricity and Gas (Market Integrty and Transparency) (Enforcement Etc.) Regulations 2013. Available online: https://www.ofgem.gov.uk/sites/de-fault/files/docs/2020/09/final_notice_signed_24_august_2020.pdf (accessed on 22 November 2021).
- 75. Ofgem Requires InterGen to Pay £37m over Energy Market Abuse. Available online: https://www.ofgem.gov.uk/publications/ofgem-requires-intergen-pay-ps37m-over-energy-market-abuse (accessed on 22 November 2021).
- 76. ESB Independent Generation Trading Limited and Carrington Power Limited Agree to Pay £6 Million for Breaching Wholesale Energy Market Regulations. Available online: https://www.ofgem.gov.uk/publications/esb-independent-generation-tradinglimited-and-carrington-power-limited-agree-pay-ps6-million-breaching-wholesale-energy-market-regulations?utm_source=Twitter&utm_medium=ESB_Carrington_Power_Pay_6m&utm_term=&utm_content=ESB_Carrington_Power_Pay_6m_PR&utm_campaign=ESB_Carrington_Power_Pay_6m (accessed on 19 November 2021).
- 77. Alban, T. Generators Fined £6 Million for Faulty Supply Stats. Available online: https://theenergyst.com/generators-fined-6-million-for-faulty-supply-stats/ (accessed on 19 November 2021).
- 78. Allen, F.; Gale, D. Stock-Price Manipulation. Rev. Financ. Stud. 1992, 5, 503–529. https://doi.org/10.1093/rfs/5.3.503.
- 79. Huang, Y.C.; Chen, R.C.Y.; Cheng, Y.J. Stock Manipulation and Its Impact on Market Quality; National Kaohsiung First University of Science and Technology: Kaohsiung, Taiwan, 2005.
- Liu, Q.; Wong, I.; An, Y.; Zhang, J. Asymmetric Information and Volatility Forecasting in Commodity Futures Markets. *Pac.-Basin Financ. J.* 2014, 26, 79–97. https://doi.org/10.1016/j.pacfin.2013.10.007.
- 81. Wang, Q.; Xu, W.; Huang, X.; Yang, K. Enhancing Intraday Stock Price Manipulation Detection by Leveraging Recurrent Neural Networks with Ensemble Learning. *Neurocomputing* **2019**, *347*, 46–58. https://doi.org/10.1016/j.neucom.2019.03.006.
- 82. Gerace, D.; Chew, C.; Whittaker, C.; Mazzola, P. Stock Market Manipulation on the Hong Kong Stock Exchange. *Australas. Account. Bus. Financ. J.* **2014**, *8*, 105–140. https://doi.org/10.14453/aabfj.v8i4.7.
- 83. Aggarwal, R.K.; Wu, G. Stock Market Manipulations. J. Bus. 2006, 79, 1915–1953. https://doi.org/10.1086/503652.
- 84. Aggarwal, R.K.; Wu, G. Stock Market Manipulation Theory and Evidence. SSRN J. 2003. https://doi.org/10.2139/ssrn.474582.
- 85. Foucault, T. Order Flow Composition and Trading Costs in a Dynamic Limit Order Market. J. Financ. Mark. 1999, 36, 99–134.
- 86. Zhihui, L.; Jin, W.; Mengyu, L. A Study on China's Stock Market Manipulation's Effects on Market Liquidity: Based on Closing Price Manipulation Behavior's Identification and Monitoring. *Financ. Res.* **2018**, 452.
- 87. Akinmade, B.; Adedoyin, F.F.; Bekun, F.V. The Impact of Stock Market Manipulation on Nigeria's Economic Performance. *Econ. Struct.* **2020**, *9*, 52. https://doi.org/10.1186/s40008-020-00226-0.
- 88. Amihud, Y. Illiquidity and Stock Returns: Cross-Section and Time-Series Effects. J. Financ. Mark. 2002, 50, 31–56.
- Shen, C.; Hu, D. Early Warning Indicators of Price Manipulation in the Chinese Stock Market. *Sci. Technol. Eng.* 2011, *11*, 912–915.
- 90. Mao, C. The Research on Liquidity Risk Measurement in Stock Market. Ph.D. Thesis, Hunan University, Changsha, China.
- 91. Ying, Z. Research on Chinese Stock Market liquidity. Secur. Mark. Guide 2001, 7, 63-63. https://doi.org/10.3969/j.issn.1005-1589.2001.07.014.
- 92. Uppal, J.Y.; Mangla, I.U. Market Volatility, Manipulation, and Regulatory Response: A Comparative Study of Bombay and Karachi Stock Markets. *Pak. Dev. Rev.* 2006, 45, 1071–1083. https://doi.org/10.30541/v45i4IIpp.1071-1083.
- 93. Lu, R.; Chen, X. Research on the market performance of stock manipulation and its discrimination. *Secur. Mark. Guide* 2009, *8*,65-72. doi:CNKI:SUN:ZQDB.0.2009-04-013.
- 94. Comerton-Forde, C.; Putniņš, T.J. Measuring Closing Price Manipulation. J. Financ. Intermediation 2011, 20, 135–158. https://doi.org/10.1016/j.jfi.2010.03.003.
- 95. Deveikyte, J.; Geman, H.; Piccari, C.; Provetti, A. A Sentiment Analysis Approach to the Prediction of Market Volatility. *arXiv* **2020**, arXiv:2012.05906.
- 96. Tetlock, P.C.; Saar-Tsechansky, M.; Macskassy, S. More Than Words: Quantifying Language to Measure Firms' Fundamentals. *J. Financ.* **2008**, *63*, 31. https://doi.org/10.1111/j.1540-6261.2008.01362.x.
- 97. Gan, B.; Alexeev, V.; Bird, R.; Yeung, D. Sensitivity to Sentiment: News vs Social Media. *Int. Rev. Financ. Anal.* 2020, 67, 101390. https://doi.org/10.1016/j.irfa.2019.101390.

- 98. Groß-Klußmann, A.; Hautsch, N. When Machines Read the News: Using Automated Text Analytics to Quantify High Frequency News-Implied Market Reactions. J. Empir. Financ. 2011, 18, 321–340. https://doi.org/10.1016/j.jempfin.2010.11.009.
- 99. Baker, M.; Wurgler, J. Investor Sentiment and the Cross-Section of Stock Returns. J. Financ. 2006, 61, 1645-1680. https://doi.org/10.1111/j.1540-6261.2006.00885.x.
- Glasserman, P.; Mamaysky, H. Does Unusual News Forecast Market Stress? J. Financ. Quant. Anal. 2019, 54, 1937–1974. https://doi.org/10.1017/S0022109019000127.
- Huang, D.; Tu, J.; Jiang, F.; Zhou, G. Investor Sentiment Aligned: A Powerful Predictor of Stock Returns. *Rev. Financ. Stud.* 2015, 28, 791–837. https://doi.org/10.1093/RFS/HHU080.
- Gabrovšek, P.; Aleksovski, D.; Mozetič, I.; Grčar, M. Twitter Sentiment around the Earnings Announcement Events. *PLoS ONE* 2017, 12, e0173151. https://doi.org/10.1371/journal.pone.0173151.
- 103. Pinczynski, M.; Kasperowicz, R. Overview of Electricity Market Monitoring. *Econ. Sociol.* 2016, 9, 153–167. https://doi.org/10.14254/2071-789X.2016/9-4/9.
- 104. Kaplow, L. Market Definition, Market Power. Int. J. Ind. Organ. 2015, 43, 148–161. https://doi.org/10.1016/j.ijindorg.2015.05.001.
- Rahimi, A.F.; Sheffrin, A.Y. Effective Market Monitoring in Deregulated Electricity Markets. *IEEE Trans. Power Syst.* 2003, 18, 486–493. https://doi.org/10.1109/TPWRS.2003.810680.
- 106. Kaiser, G.E. The Guide to Energy Market Competition. Available online: https://media.crai.com/wp-content/up-loads/2020/09/16164449/Economic-Evidence-of-Market-Power-and-Market-Manipulation-in-Energy-Market.pdf (accessed on 11 April 2022).
- 107. Adib, P.; Hurlbut, D. Market Power and Market Monitoring. In *Competitive Electricity Markets*; Elsevier: Amsterdam, The Netherlands, 2008; pp. 267–296. ISBN 978-0-08-047172-3.
- Cardell, J.B.; Hitt, C.C.; Hogan, W.W.; Shogren, J.F.; Smulders, S. Market Power and Strategic Interaction in Electricity Networks. *Resour. Energy Econ.* 1997, 19, 109–137. https://doi.org/10.1016/S0928-7655(97)00006-7.
- 109. Market Based Rate Standards. Available online: https://www.ferc.gov/industries-data/natural-gas/intrastate-transportation/market-based-rate-standards (accessed on 28 February 2022).
- Twomey, P.; Green, R.; Neuhoff, K.; Newbery, D. A Review of the Monitoring of Market Power: The Possible Roles of TSOs in Monitoring for Market Power Issues in Congested Transmission Systems. *Camb. Work. Pap. Econ.* 2005, Available online: https://ceepr.mit.edu/wp-content/uploads/2023/02/2005-002.pdf (accessed on 28 February 2022).
- 111. He, A.; Tan, H.; Jin, Z. The Control of Market Power in Electricity Market. Available online: https://mp.weixin.qq.com/s/FkMtCee_xgS5R-u3PaTHg (accessed on 28 February 2022).
- 112. He, J.; Zhao, W.; Huang, H.; Wang, M.; Zhao, P. The Evaluation System of Power Market Monitoring Based on AHP and the Entropy Method. *IOP Conf. Ser. Earth Environ. Sci.* 2021, *831*, 12027. https://doi.org/10.1088/1755-1315/831/1/012027.
- 113. Swinand, G.; Scully, D.; Ffoulkes, S.; Kessler, B. Modeling EU Electricity Market Competition Using the Residual Supply Index. *Electr. J.* **2010**, *23*, 41–50. https://doi.org/10.1016/j.tej.2010.09.016.
- 114. Bataille, M.; Bodnar, O.; Steinmetz, A.; Thorwarth, S. Screening Instruments for Monitoring Market Power—The Return on Withholding Capacity Index (RWC). *Energy Econ.* **2019**, *81*, 227–237. https://doi.org/10.1016/j.eneco.2019.03.011.
- Bataille, M.; Steinmetz, A.; Thorwarth, S. Screening Instruments for Monitoring Market Power in Wholesale Electricity Markets Lessons from Applications in Germany. SSRN J. 2014. https://doi.org/10.2139/ssrn.2475943.
- 116. Baker, J.; Timothy, F.B. Empirical Methods of Identifying and Measuring Market Power. Antitrust Law J. 1992, 61, 15.
- 117. Chang, Y. The New Electricity Market of Singapore: Regulatory Framework, Market Power and Competition. *Energy Policy* **2007**, *35*, 403–412. https://doi.org/10.1016/j.enpol.2005.11.036.
- Kamiński, J. The Development of Market Power in the Polish Power Generation Sector: A 10-Year Perspective. *Energy Policy* 2012, 42, 136–147. https://doi.org/10.1016/j.enpol.2011.11.057.
- 119. Helman, U. Market Power Monitoring and Mitigation in the US Wholesale Power Markets. *Energy* 2006, 31, 877–904. https://doi.org/10.1016/j.energy.2005.05.011.
- 120. Biggar, D. *The Theory and Practice of the Exercise of Market Power in the Australian Nem;* Australian Competition and Consumer Commission: Melbourne, Australia, 2002; p. 75.
- Visudhiphan, P.; Ilic, M.D. Dependence of Generation Market Power on the Demand/Supply Ratio: Analysis and Modeling. In Proceedings of the 2000 IEEE Power Engineering Society Winter Meeting. Conference Proceedings (Cat. No.00CH37077), Singapore, 23–27 January 2000; Volume 2, pp. 1115–1122.
- 122. Prabhakar Karthikeyan, S.; Jacob Raglend, I.; Kothari, D.P. A Review on Market Power in Deregulated Electricity Market. *Int. J. Electr. Power Energy Syst.* 2013, 48, 139–147. https://doi.org/10.1016/j.ijepes.2012.11.024.
- 123. Dun-nan, L.; Rui-qing, L.; Guang-yu, H.; Xue-qing, C. A Market Analysis and Evaluating System for Surveillance of Electricity Market. In Proceedings of the 2004 IEEE International Conference on Electric Utility Deregulation, Restructuring and Power Technologies, Hong Kong, China, 5–8 April 2004; pp. 556–561.
- 124. Cramer, D.L.; Heuser, W.L. Variations in the Definitions of the Degrees of Competition. *Am. J. Econ. Sociol.* **1960**, *19*, 383–383. https://doi.org/10.1111/j.1536-7150.1960.tb00405.x.
- 125. Jin, L.; Chen, C.; Yu, J.; Wang, X.; Miao, Q. The Compliance Risk Assessment of Electric Power Market Based on Bayesian Comprehensive Evaluation Model. *IOP Conf. Ser. Earth Environ. Sci.* 2021, 647, 012150. https://doi.org/10.1088/1755-1315/647/1/012150.

- 126. Kasperowicz, R.; Pinczyński, M.; Tiwari, A.; Nawrot, Ł. Reengineering of Electricity Market Monitoring. *Econ. Sociol.* **2017**, *10*, 175–188. https://doi.org/10.14254/2071-789X.2017/10-4/14.
- 127. Luis, A. A. Preparing for the Regulatory Challenges of the 21st Century. Available online: https://www.sec.gov/news/speech/preparing-for-regulatory-challenges-of-21st-century.html (accessed on 16 April 2022).
- 128. Technological Challenges to Effective Market Surveillance Issues and Regulatory Tools. Available online: https://www.iosco.org/library/pubdocs/pdf/IOSCOPD389.pdf (accessed on 2 April 2022).
- 129. Fletcher, G.-G.S. Deterring Algorithmic Manipulation. Vanderbilt Law Rev. 2021, 74, 259.
- 130. Wolak, F.A. Lessons from International Experience with Electricity Market Monitoring. World Bank 2005, 3692, 23.
- 131. Huang, X.; Guo, G. Analysis of Market Power in Iiberalized Electricity Markets. *China Power* 2003, *6*, 35–40. https://doi.org/10.3969/j.issn.1004-9649.2003.06.009.
- 132. Michael, L. What Are the Biggest Regulatory Challenges Facing the Energy Sector? Available online: https://www.energynetworks.com.au/news/energy-insider/what-are-the-biggest-regulatory-challenges-facing-the-energy-sector/ (accessed on 19 April 2022).

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.